Author's response to reviews

Title: Biweekly cetuximab in combination with FOLFOX-4 in the first-line treatment of wild-type KRAS metastatic colorectal cancer: final results of a phase II, open-label, clinical trial (OPTIMIX-ACROSS Study)

Authors:

Julen Fernandez-Plana (julenfernandez@mutuaterrassa.cat)
Carlos Pericay (cpericay@gmail.com)
Guillermo Quintero (gquinteroaldana@gmail.com)
Vicente Alonso (valonsoo@salud.aragon.es)
Antonieta Salud (asaluds@hotmail.com)
Miguel Mendez (mmendez.hmtl@salud.madrid.org)
Mercedes Salgado (mercedes.salgado.fernandez@sergas.es)
Eugeni Saigi (esaigi@tauli.cat)
Luis Cirera (lcirera@mutuaterrassa.es)

Version: 2
Date: 14 May 2014

Author's response to reviews:

Dear Editor,

A new manuscript version of the manuscript has been uploaded including:

1) A title page in the manuscript document
2) Page numbers of the document
3) Line numbers in all the main document

Thanks for your help.

Regards,
Luis Cirera